

Identifying Hydro-Sensitive Coral δ 18 O Records for Improved High-Resolution Temperature and Salinity Reconstructions

D. Thompson, J. Conroy, B. Konecky, S. Stevenson, K. Delong, N. Mckay, E. Reed, L. Jonkers, Matthieu Carré

▶ To cite this version:

D. Thompson, J. Conroy, B. Konecky, S. Stevenson, K. Delong, et al.. Identifying Hydro-Sensitive Coral $\delta 18$ O Records for Improved High-Resolution Temperature and Salinity Reconstructions. Geophysical Research Letters, 2022, 49 (9), pp.e2021GL096153. 10.1029/2021gl096153. hal-03726613

HAL Id: hal-03726613 https://hal.science/hal-03726613v1

Submitted on 16 Aug2022

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Geophysical Research Letters[•]

RESEARCH LETTER

10.1029/2021GL096153

Thompson and Conroy contributed equally to this work.

Key Points:

- Seawater $\delta^{18}O$ contributes up to 89% of the coral $\delta^{18}O$ variability, providing critical information about salinity variability and trends
- The inferred salinity contribution to coral δ^{18} O values is highly dependent on the salinity- δ^{18} O linear regression slope
- Interpretation of coral 8¹⁸O values should account for both temperature and hydrological influences, which may oppose one another

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

D. M. Thompson, thompsod@email.arizona.edu

Citation:

Thompson, D. M., Conroy, J. L., Konecky, B. L., Stevenson, S., DeLong, K. L., McKay, N., et al. (2022). Identifying hydro-sensitive coral δ¹⁸O records for improved highresolution temperature and salinity reconstructions. *Geophysical Research Letters*, 49, e2021GL096153. https://doi. org/10.1029/2021GL096153

Received 21 SEP 2021 Accepted 20 APR 2022

Author Contributions:

Conceptualization: D. M. Thompson, J. L. Conroy, B. L. Konecky, S. Stevenson, K. L. DeLong, N. McKay Data curation: D. M. Thompson, J. L. Conroy, B. L. Konecky, S. Stevenson, K. L. DeLong, N. McKay, E. Reed, L. Jonkers, M. Carré Formal analysis: D. M. Thompson, J. L. Conroy, S. Stevenson Funding acquisition: D. M. Thompson, J. L. Conroy, B. L. Konecky, S. Stevenson, K. L. DeLong, E. Reed, L. Jonkers Investigation: D. M. Thompson, J. L. Conroy, S. Stevenson

© 2022. American Geophysical Union. All Rights Reserved.

Identifying Hydro-Sensitive Coral δ^{18} O Records for Improved High-Resolution Temperature and Salinity Reconstructions

D. M. Thompson¹, J. L. Conroy², B. L. Konecky³, S. Stevenson⁴, K. L. DeLong⁵, N. McKay⁶, E. Reed¹, L. Jonkers⁷, and M. Carré⁸

¹Department of Geosciences, University of Arizona, Tucson, AZ, USA, ²Department of Geology and Department of Plant Biology, University of Illinois, Urbana-Champaign, IL, USA, ³Department of Earth and Planetary Sciences, Washington University, Saint Louis, MO, USA, ⁴Bren School of Environmental Science & Management, University of California, Santa Barbara, CA, USA, ⁵Department of Geography and Anthropology, Coastal Studies Institute, Louisiana State University, Baton Rouge, LA, USA, ⁶School of Earth and Sustainability, Northern Arizona University, Flagstaff, AZ, USA, ⁷MARUM Center for Marine Environmental Sciences, Bremen University, Bremen, Germany, ⁸IPSL-LOCEAN Laboratory, Sorbonne Universités (UPMC)-CNRS-IRD-MNHN, Paris, France

Abstract Stable oxygen isotopic ratios in corals ($\delta^{18}O_{coral}$) are commonly utilized to reconstruct climate variability beyond the limit of instrumental observations. These measurements provide constraints on past seawater temperature, due to the thermodynamics of isotopic fractionation, but also past salinity, as both salinity and seawater $\delta^{18}O(\delta^{18}O_{sw})$ are similarly affected by precipitation/evaporation, advection, and other processes. We use historical observations, isotope-enabled model simulations, and the PAGES Iso2k database to assess the potential of $\delta^{18}O_{coral}$ to provide information on past salinity. Using "pseudocorals" to represent $\delta^{18}O_{coral}$ as a function of observed or simulated temperature and salinity/ $\delta^{18}O_{sw}$, we find that $\delta^{18}O_{sw}$ -salinity relationship influences the inferred salinity variability, corals from these sites could provide valuable $\delta^{18}O_{sw}$ reconstructions. Coordinated in situ monitoring of salinity and $\delta^{18}O_{sw}$ is vital for improving estimates of hydroclimatic change.

Plain Language Summary Seawater salinity provides valuable information about the global water cycle, including changes in precipitation, evaporation, and ocean currents. However, scientists' understanding of recent salinity change is hampered by the limited coverage of salinity measurements. Tropical corals incorporate environmental information in their skeletons as they grow, providing indirect evidence of past ocean salinity. Quantitatively estimating salinity variations requires accounting both for temperature- and salinity-driven influences on the stable oxygen isotope composition (δ^{18} O) of the coral skeleton, as well as changes in the relationship between salinity and seawater δ^{18} O. Through comparison with instrumental records and climate models, we find that corals in the western Pacific Ocean and eastern Indian Ocean are most likely to provide reliable salinity information across seasons, years, and decades. Finally, we find that direct seawater δ^{18} O observations are critical to improve the salinity information obtained from corals.

1. Introduction

Tropical sea surface temperature (SST) and sea surface salinity (SSS), variables driven by large-scale oceanic-atmospheric interactions, are recorded in the stable oxygen isotopic ratio of coral skeletal aragonite ($\delta^{18}O_{coral}$). Coral $\delta^{18}O$ values thus preserve key information on past ocean-atmosphere variability on interannual to centennial timescales, such as the interannual El Niño-Southern Oscillation (ENSO) in the tropical Indo-Pacific (e.g., Cobb et al., 2013; Cole and Fairbanks, 1990; Grothe et al., 2020; McGregor et al., 2013). A sound understanding of how $\delta^{18}O_{coral}$ values record changes in SST and SSS is vital for understanding past climate variability. Coral $\delta^{18}O$ values co-vary positively with the $\delta^{18}O$ of the seawater from which the skeleton forms ($\delta^{18}O_{sw}$), and co-vary negatively with the water temperature at the time of skeletal precipitation due to the thermodynamics of isotope fractionation (Emiliani et al., 1978; Epstein et al., 1953; Weber & Woodhead, 1972). Further, SSS has a strong positive relationship with $\delta^{18}O_{sw}$ in the tropical-subtropical oceans, as precipitation leads to freshening and lower $\delta^{18}O_{sw}$ values, whereas evaporation leads to increased salinity and higher $\delta^{18}O_{sw}$ values (Craig & Gordon, 1965; Fairbanks et al., 1997; LeGrande & Schmidt, 2006). Ocean currents and upwelling advect surface and subsurface waters with particular SSS- $\delta^{18}O_{sw}$ signatures imparted by the precipitation-evaporation (P-E) balance of



Methodology: D. M. Thompson, J. L. Conroy Project Administration: D. M. Thompson, J. L. Conroy, B. L. Konecky, N. McKay Validation: D. M. Thompson, J. L. Conroy Visualization: D. M. Thompson, J. L. Conroy, S. Stevenson Writing - original draft: D. M. Thompson, J. L. Conroy, B. L. Konecky, S. Stevenson, K. L. DeLong Writing - review & editing: D. M. Thompson, J. L. Conroy, B. L. Konecky, S. Stevenson, K. L. DeLong, N. McKay, E. Reed, L. Jonkers, M. Carré

the source region, resulting in further regional-scale SSS- $\delta^{18}O_{sw}$ coherence (Stevenson et al., 2018) that may not covary with local SST (Watanabe et al., 2019).

SST and SSS typically contribute constructively to $\delta^{18}O_{coral}$ values in the tropical-subtropical oceans. Where warmer and wetter conditions co-occur, the $\delta^{18}O_{coral}$ signal is additive, since precipitation results in lower salinity and $\delta^{18}O_{sw}$, and warmer temperatures also lower $\delta^{18}O_{coral}$ values. For example, the co-occurrence of warm, wet conditions in the central Pacific region during El Niño events amplifies the signal in $\delta^{18}O_{coral}$, providing clearly detectable and measurable ENSO anomalies over the Holocene (e.g., Cobb et al., 2013; Cole and Fairbanks, 1990; Grothe et al., 2020; McGregor et al., 2013). However, in other locations, such as the eastern equatorial Pacific, seasonal temperature and precipitation extremes may lag each other by several months or may interfere destructively, adding uncertainty to the interpretation of $\delta^{18}O_{coral}$ values (Cahyarini et al., 2008; Russon et al., 2013). Paired coral Sr/Ca and $\delta^{18}O$ measurements provide a means of determining the contribution of SST and SSS/ $\delta^{18}O_{sw}$ to $\delta^{18}O_{coral}$ values since Sr/Ca is primarily sensitive to SST (Corrège, 2006). However, most published $\delta^{18}O_{coral}$ records to date lack paired Sr/Ca data (Carré et al., 2021; Konecky et al., 2020).

An extensive body of work has focused on isolating the SST and ENSO signals in $\delta^{18}O_{coral}$ (e.g., Cobb et al., 2013; Cole and Fairbanks, 1990; Hereid et al., 2013; Lawman et al., 2020; Tierney et al., 2015), but less attention has been paid to the $\delta^{18}O_{sw}$ signal, despite its local significance at some coral sites (e.g., Cahyarini et al., 2008; Cahyarini et al., 2014; Dassié et al., 2018; Gorman et al., 2012; Murty et al., 2018). An inquiry into the role of $\delta^{18}O_{sw}$ is especially timely, as trends in SSS since the mid-20th century (when SSS measurements became more common) echo trends in atmospheric P-E and are a fingerprint of anthropogenic hydroclimatic change (Durack et al., 2012).

Given the limited temporal range of SSS measurements, paleo- $\delta^{18}O_{sw}$ estimates from corals have potential value as proxies for past SSS. Although robust paleo-SSS reconstructions from corals may be limited by uncertainties in the slope of the $\delta^{18}O_{sw}$ -SSS relationship, a systematic investigation into the potential of coral records to provide robust SSS information has not yet been performed. In this paper, we comprehensively explore the relationship between $\delta^{18}O_{coral}$ and SSS using synthetic $\delta^{18}O_{coral}$ records, or "pseudocorals" (Thompson et al., 2011), modeled from observed and simulated temperature, SSS, and $\delta^{18}O_{sw}$. This approach permits a detailed investigation of the relative role of temperature and SSS/ $\delta^{18}O_{sw}$ in modern $\delta^{18}O_{coral}$ interannual to multidecadal variability across the tropics, and identifies optimal sites for reconstructing SSS. Applying this approach to coral sites in the new PAGES Iso2k database (Konecky et al., 2020) with records spanning the instrumental period, we identify hydro-sensitive $\delta^{18}O_{coral}$ records that can provide critical salinity information.

2. Methods

2.1. Identification of Hydro-Sensitive Coral Sites From Observed or Simulated SST and SSS/ $\delta^{18}O_{sw}$

To identify sites where $\delta^{18}O_{sw}$ contributes strongly to $\delta^{18}O_{coral}$ variability, $\delta^{18}O_{coral}$ was first predicted from historical or simulated SST and $\delta^{18}O_{sw}$ /SSS over the 1950–2009 period using a coral proxy system model (PSM, Thompson et al., 2011):

$$\delta^{18}O_{\text{coral}} = a_1 \text{SST} + b\delta^{18}O_{\text{sw}} \tag{1a}$$

$$\delta^{18}O_{\text{coral}} = a_1 \text{SST} + a_2 \text{SSSb} + \varepsilon \tag{1b}$$

where the coefficient a_1 was set to $-0.22 \text{ %c/}^{\circ}$ C based on empirical and theoretical estimates of the $\delta^{18}O_{coral}$ -SST relationship (Epstein et al., 1953; Evans et al., 2000; Lough, 2004). In model output, where $\delta^{18}O_{sw}$ estimates were available, $\delta^{18}O_{coral}$ was calculated directly from $\delta^{18}O_{sw}$ using the relationship between the Vienna Pee Dee Belemnite (VPDB) and Vienna Standard Mean Ocean Water reference standards for $\delta^{18}O_{carbonate}$ and $\delta^{18}O_{sw}$, respectively (b = 0.97002; Equation 1a). SSS was used in place of $\delta^{18}O_{sw}$ for estimates over the historical period, due to the paucity of *in situ* $\delta^{18}O_{sw}$ observations. In these cases, a_2 was defined at each site by the regional slopes of the $\delta^{18}O_{sw}$ -SSS relationship (LeGrande & Schmidt, 2006), and the resulting $\delta^{18}O_{sw}$ was converted to VPDB (Equation 1b).

We identified sites where $\delta^{18}O_{sw}$ contributes strongly to $\delta^{18}O_{coral}$ variability using two metrics derived from this forward-modeled coral $\delta^{18}O_{coral}$. The first metric, the F_{sw} ("fraction seawater") statistic, quantifies the fraction



of total $\delta^{18}O_{coral}$ variance driven by both $\delta^{18}O_{sw}$ variance and the covariance of $\delta^{18}O_{sw}$ with SST (F_{cov} ; Russon et al., 2013). The second metric, F_{cov} ("fraction covariance"), assesses the fraction of $\delta^{18}O_{coral}$ due solely to SST- $\delta^{18}O_{sw}$ covariance (Russon et al., 2013). As in Equations 1a, 1b, $\delta^{18}O_{sw}$ was replaced by SSS, using the regional $\delta^{18}O_{sw}$ - SSS relationship in the absence of sufficient $\delta^{18}O_{sw}$ observations over the historical period.

$$F_{\rm cov} = \frac{2R_{\rm SST-coral}\,\operatorname{cov}\left({\rm SST}, \delta^{18}{\rm O}_{\rm sw}\right)}{\operatorname{var}\left(\delta^{18}{\rm O}_{\rm coral}\right)}\tag{2}$$

$$F_{\rm sw} = \frac{\operatorname{var}\left(\delta^{18}O_{\rm sw}\right) + 2R_{\rm SST-coral}\operatorname{cov}\left(\mathrm{SST}, \delta^{18}O_{\rm sw}\right)}{\operatorname{var}\left(\delta^{18}O_{\rm coral}\right)}$$
(3)

where $R_{\text{SST-coral}}$ is the slope of the $\delta^{18}O_{\text{coral}}$ -SST relationship of $-0.22 \text{ %c/}^{\circ}$ C. The F_{cov} and F_{sw} values can range from -1 to 1. For F_{cov} , negative values occur at sites where there is a positive covariance between SST and SSS (and thus a positive SST-SSS correlation). Correlation coefficients between annual anomalies of SST and SSS were also calculated to reveal the direction and significance of this relationship at each coral site. In turn, F_{sw} values greater than zero indicate a fraction of $\delta^{18}O_{\text{coral}}$ variance stems from $\delta^{18}O_{\text{sw}}$ variance and its covariance with SST. Negative values are rare, but arise if there is positive covariance between SST and $\delta^{18}O_{\text{sw}}/\text{SSS}$ (i.e., $r_{\text{SST-SSS}}$ is positive and the F_{cov} term is negative) that surpasses that of $\delta^{18}O_{\text{sw}}$ variability. In any situation where there is positive covariance between SST and $\delta^{18}O_{\text{sw}}/\text{SSS}$, SST and SSS destructively interfere in $\delta^{18}O_{\text{coral}}$ and $\delta^{18}O_{\text{coral}}$ variance is dampened relative to that of SST or SSS. For example, warmer SST coupled with higher $\delta^{18}O_{\text{sw}}$ and salinity will reduce $\delta^{18}O_{\text{coral}}$ variance in regions with a Mediterranean climate (i.e., dry summer, wet winter) (Russon et al., 2013), which reduces the signal-to-noise ratio and limits the ability to detect and analyze the paleoclimate signal.

2.2. Sensitivity Analyses (Datasets, Coefficients, and Timescales of Variability)

A series of sensitivity tests were conducted to assess the impact of parametric and observational uncertainties on predicted F_{sw} values and identify whether the inferred salinity signal varies across timescales (from interannual to multidecadal). First, multiple SST and SSS data sources were assessed, including HadISST v1.1 SST, OISST v2.0 SST, SODA v2.2.4 SST/SSS, Delcroix (tropical Pacific) SSS, IAP SSS, and ORAS v5 (ensemble members 0–4) SSS (Behringer & Xue, 2004; Carton & Giese, 2008; Cheng et al., 2020; Delcroix et al., 2011; Giese & Ray, 2011; Rayner et al., 2003; Reynolds et al., 2002; Zuo et al., 2019). Most of the SST and SSS data products are interpolated from observations to be globally complete whereas SODA and ORAS are reanalysis products that assimilate historical observations into a numerical model. HadISST v1.1 and SODA v2.2.4 are presented in Figure 2, selected for their high spatial resolution; however, we do not advocate here for any data set as superior to the others, rather, sensitivity tests with different data sets are included in Figure S2 in Supporting Information S1. All analyses were performed on April-March annual anomalies in MATLAB, with anomalies calculated relative to the long-term monthly mean (i.e., removing the seasonal cycle) over the 1950–2009 period prior to averaging over the tropical year. The only exception was the ''seasonal to interannual'' anomaly analysis, Figure S8 in Supporting Information S1, which was performed on anomalies calculated by removing the full timeseries mean from annual tropical averages.

We then assessed the impact of uncertainties in the $\delta^{18}O_{sw}$ -SSS relationship in two key ways. First, we model $\delta^{18}O_{coral}$ directly from $\delta^{18}O_{sw}$, rather than converting SSS data to $\delta^{18}O_{sw}$ using a proportionality constant, thus bypassing this uncertainty in SSS-based estimates. Although no spatiotemporally complete $\delta^{18}O_{sw}$ reanalysis product exists to date, the isotope-enabled Regional Ocean Modeling System (isoROMS; Stevenson et al., 2015, 2018) provides the closest approximation, with a nominal resolution of approximately 50 km spanning the Pacific basin. We also compare isoROMS results with those from $\delta^{18}O_{sw}$ simulated by an isotope-enabled simulation of the Community Earth System Model (iCESM) (Nusbaumer et al., 2017; Zhang et al., 2017; Zhu et al., 2017). We note that these comparisons were performed over the 1965–2005 period, due to the presence of minor model drift in the early portion of the isoROMS simulation.

We then assessed the dependence of the predicted F_{sw} values on coefficient a_2 . At each site, we randomly perturbed the a_2 coefficient using 1,000 random draws of a Gaussian distribution of a_2 coefficients, centered on the expected value taken from Legrande and Schmidt (2006). The 1 sigma uncertainty of the a_2 coefficient was set to 0.1%/PSU based on the standard deviation of the $\delta^{18}O_{sw}$ -SSS relationship across the limited network of

long-term seawater monitoring sites (Conroy et al., 2017). We next applied a more conservative a_2 uncertainty of 0.2‰/PSU (Figure S4 in Supporting Information S1) and identified the impact of this parametric uncertainty on the inferred $\delta^{18}O_{coral}$ trends for the late 20th century (Figure S5 in Supporting Information S1). We then tested spatially modified slopes based on the east-to-west differences in the $\delta^{18}O_{sw}$ -SSS slope identified by Conroy et al. (2017) and compared the F_{sw} values with those obtained with the Pacific-wide slopes of LeGrande and Schmidt (2006) (Figure S6 in Supporting Information S1).

Finally, we tested the impact of timescales of SSS- $\delta^{18}O_{sw}$ variability on the F_{sw} estimates to assess the utility of $\delta^{18}O_{coral}$ for reconstructing salinity variability across timescales. Each analysis was performed following the methods described above, but with four additional sensitivity tests (Figures S8-S12 in Supporting Information S1): (a) linear detending, (b) removing the seasonal cycle prior to calculating the annual tropical average ("interannual") versus removing the full time-series mean after calculating the annual tropical average ("seasonal to interannual"), and using the idealfilter function in MATLAB to isolate variability in (c) the ENSO band using a 2–7 years bandpass filter and in the (d) the decadal-multidecadal band using a 10–40 years bandpass filter.

2.3. Application to Coral δ^{18} O Records From the Iso2k Database

Coral δ^{18} O records (N = 88) that contain modern (>1950 CE) values were extracted from the Iso2k database v1.0.0 (Konecky and McKay, 2020; Konecky et al., 2020). This version of the Iso2k database contains coral δ^{18} O records published before 4 May 2018. As with the observational and model analyses described above, all monthly δ^{18} O_{coral} data were averaged into tropical (April-March) years before analysis (Table S1, Figure S1 in Supporting Information S1).

We first used the F_{sw} framework to identify sites expected to have a strong salinity signal in annual-averaged $\delta^{18}O_{coral}$ that is undamped by SST variability. This approach allows us to identify targets for future paleosalinity reconstruction. Partial correlation coefficients between annual averages of SSS and Iso2k $\delta^{18}O_{coral}$ from 1950 to 2009, after removing the influence of SST, were also assessed to reveal which actual records have a significant relationship with historical SSS. Finally, we calculated the slope of the temporal trend in $\delta^{18}O_{coral}$ records from 1900 (or the beginning of the record, if later) to 2009 at sites where $F_{sw} > 0.5$ and the coral record has a significant, positive partial correlation with SSS ("hydro-sensitive coral sites"). We exclude sites with <75 years of data during this time interval from the trend analysis.

3. Results

Pseudocoral F_{sw} values are positive in a majority (89%) of coral sites, and above 0.5 in 42% of the sites, reaching 0.89 in two western Pacific sites. A negative relationship between annual SSS and SST (Figures 1 and 2c) is observed at most sites, although the correlation is often statistically insignificant (p > 0.05). Ten pseudocorals (6 sites) have an F_{sw} value ≤ 0 and a positive SSS-SST relationship. These observations are broadly consistent when using different SSS and SST products (Figures S2-S3 in Supporting Information S1), but with notable differences in the Indian Ocean. Although some differences are also observed in the southeast Pacific and south Atlantic, no coral records have been collected from these regions. The value of the a_2 coefficient contributes strongly to the magnitude of the F_{sw} value, with lower a_2 values producing lower F_{sw} values and vice versa (Figure 3, Figure S4 in Supporting Information S1). Despite its importance in determining the absolute magnitude of F_{sw} (Figure 3, S4, S6 in Supporting Information S1), the a_2 coefficient contributes little uncertainty to the temporal trends of pseudocoral δ^{18} O values (assuming that a_2 is stable through time, Figure S7 in Supporting Information S1) or to the selection of hydro-sensitive sites (although fewer sites display high F_{sw} as a_2 decreases; Figure 3). Geographically, many of the high F_{sw} values are found in the Indo-Pacific Warm Pool (IPWP, i.e., the western tropical Pacific and eastern Indian Ocean; Figure 2a).

Many modern $\delta^{18}O_{coral}$ records from the Indo-Pacific Warm Pool also have a significant, positive relationship with SSS after removing the influence of SST (r_p , Figure 2d). Across all sites, F_{sw} and r_p display a significant positive correlation (r = 0.26, N = 61, p < 0.05), and mean r_p is significantly higher at sites with $F_{sw} > 0.5$ (t = 2.55, df = 86, p = 0.0126; Figure S5 in Supporting Information S1). There is also a greater spread in r_p at sites where $F_{sw} < 0.5$, with the lower quartile extending into negative values (Figure S5 in Supporting Information S1). Considering the $\delta^{18}O_{coral}$ temporal trends in the 21 records from hydro-sensitive coral sites, most (17/21)





Figure 1. Scatterplot of correlation coefficients of HadISST v1.1 and SODA v2.2.4 sea surface salinity (SSS) annual anomalies versus pseudocoral F_{sw} values at grids approximating coral proxy locations, calculated over the 1950–2009 period. The area shaded blue (1) indicates sites where $\delta^{18}O_{sw}$ contributes to $\delta^{18}O_{coral}$ variance and there is a significant negative correlation between SSS and sea surface temperature (SST). The area shaded red (2) indicates sites where $\delta^{18}O_{sw}$ contributes to $\delta^{18}O_{coral}$ variance and there is no significant correlation between SSS and SST. In area (3), F_{sw} values are <0 and $\delta^{18}O_{sw}$ -SSS contributes destructively to $\delta^{18}O_{coral}$.

have a negative trend and are located in the IPWP (Figure 4). The length of these records commonly extends beyond the limits of the reliable, instrumental SSS record (i.e., >50 years).

 F_{sw} displays similar spatial patterns across seasonal and interannual timescales (Figure S8 in Supporting Information S1), and among historical data products (Figure S2 in Supporting Information S1) and isotope-enabled simulations (Figure S9 in Supporting Information S1). Interannual $\delta^{18}O_{sw}$ variability contributes strongly to high F_{sw} values in the South Pacific Convergence Zone (SPCZ) region (Figure S8b in Supporting Information S1), whereas strong negative correlations between SST and $\delta^{18}O_{sw}$ contribute to the high F_{sw} values in the central-western equatorial Pacific on seasonal timescales (Figure S8f in Supporting Information S1, and to a lesser extent, on interannual timescales (Figure S10f in Supporting Information S1). F_{sw} in the 2–7 years ENSO band and 10–40 years band shows no major differences to those calculated from unfiltered data in Figure 2 (Figures S10-S11 in Supporting Information S1). ENSO-related $\delta^{18}O_{sw}$ variability and SST- $\delta^{18}O_{sw}$ covariance contribute to F_{sw} in the central-western equatorial Pacific, whereas decadal-multidecadal $\delta^{18}O_{sw}$ variability and SST- $\delta^{18}O_{sw}$ variability of F_{sw} in the ITCZ region (and to a lesser extent the SPCZ, Figure S10 in Supporting Information S1). Finally, F_{sw} values are greater when the long-term trend is removed, particularly in the subtropics (Figure S12 in Supporting Information S1). As these analyses of temporal variability in F_{sw} used the same regional a_2 coefficients, the observed differences in F_{sw} arise from differences in SSS/ $\delta^{18}O_{sw}$ and SST variability and their covariance.

4. Discussion

Substantial information on $\delta^{18}O_{sw}$ and SSS is likely recorded in modern and fossil coral $\delta^{18}O$ values in the locations where $F_{sw} > 0.5$. High F_{sw} values ($F_{sw} > 0.5$) are found across the tropics, but many of the sites with especially high F_{sw} values ($F_{sw} > 0.6$ –0.7) are found in the western tropical Pacific and eastern Indian Ocean, in a broad region stretching from the SPCZ, across the Maritime Continent, and northward into the South China and the Philippine Seas. Across this same region, many coral $\delta^{18}O$ records also have a positive, significant correlation





Figure 2. (a) Fraction seawater (F_{sw}) in pseudocorals at coral proxy sites, calculated from HadISST v1.1 and SODA v2.2.4 sea surface salinity (SSS) as in Figure 1. Higher values indicate a greater fraction of pseudocoral δ^{18} O variance is controlled by $\delta^{18}O_{sw}$. Sites with $F_{sw} > 0$ and a significant SSS-sea surface temperature (SST) correlation are outlined in black solid lines (blue shaded region in Figure 1). Sites with $F_{sw} > 0$ and an insignificant SSS-SST correlation are outlined in dashed lines (red shaded region in Figure 1). (b) Fraction of annual coral δ^{18} O variance due to $\delta^{18}O_{sw}$ -SST covariance at each coral site, calculated over the 1950–2009 period. (c) Correlation coefficient between SSS and SST, calculated over the 1950–2009 period. (d) Partial r_p , the $\delta^{18}O_{coral}$ -SSS correlation between annual $\delta^{18}O$ in real corals from the Iso2k database (sites where annual N > 10) and SODAv2.2.4 SSS 1950–2009, after removing the influence of HadISST v1.1. Correlations with p < 0.05 are outlined in black.

with SSS from 1950 to 2009 after removing the influence of SST. Thus, this region is a compelling target for paleosalinity reconstructions.

In some sites, however, the contribution of SSS or $\delta^{18}O_{sw}$ is notably different in pseudocorals (Figure 2a) and actual corals (Figure 2d), which may arise from differences between local and large-scale marine conditions, errors in the historical salinity data, and/or non-climatic noise in the $\delta^{18}O_{coral}$ data. We therefore caution that the identification of hydro-sensitive coral sites using a pseudocoral approach is only as robust as the observed or simulated data and the PSM used to predict the $\delta^{18}O_{coral}$ response. Nevertheless, the agreement in hydro-sensitive sites among input datasets and the significant SSS signal in observed $\delta^{18}O_{coral}$ data at most sites supports the general utility of our approach.

The lack of a strong or significant correlation between annual average SSS and SST at many of these sites implies that stationary SSS-SST covariance should not be assumed for interannual and longer timescales (i.e., lower $\delta^{18}O_{coral}$ anomalies may not necessarily always imply both warmer and fresher conditions, and vice versa). SST- $\delta^{18}O_{sw}$ correlations on interannual timescales are only weakly negative across much of the tropical





Figure 3. Impact of a_2 uncertainty on F_{sw} : (a) median F_{sw} ; (b) 5th percentile, (c) 95th percentile, (d) the confidence interval (CI, difference between the 5% and 95%), calculated from 1,000 random draws of a Gaussian distribution of a_2 values, with the mean value from LeGrande and Schmidt (2006) and standard deviation of 0.1. Pseudocoral F_{sw} estimates were calculated from HadISST v1.1 and SODA v2.2.4 sea surface salinity, as in Figure 1. Coral sites are outlined in black.

Indo-Pacific Ocean (Figure S8d in Supporting Information S1), whereas a strongly negative SST- $\delta^{18}O_{sw}$ relationship plays an important role in $\delta^{18}O_{coral}$ variability in the central-western Pacific on seasonal to interannual timescales (Figures S8f, S10f in Supporting Information S1). Nevertheless, very few sites display destructive interference between SSS and SST in the form of a positive, significant correlation coefficient that confounds $\delta^{18}O_{coral}$ variability. We do note that this analysis is confined to recent decades; the SST-SSS relationship may vary greatly on millennial timescales with large geographical shifts in the location of convergence zones (LeGrande & Schmidt, 2011) leading to changes in coral response to ENSO across the Holocene (Carré et al., 2021). For



Figure 4. Slope of Iso2k $\delta^{18}O_{coral}$ z-score per year, over the 1900–2009 period for sites with F_{sw} >0.5, a $\delta^{18}O_{coral}$ -sea surface salinity partial correlation coefficient with p < 0.1, and at least 75 years of annual $\delta^{18}O_{coral}$ values from 1900 to 2009. Sites that do not meet these criteria are shaded gray. Marker size denotes the length of the Iso2k coral record. Negative $\delta^{18}O_{coral}$ trends (blue) imply local freshening, whereas positive $\delta^{18}O_{coral}$ trends (red) suggest local salinification.

example, Corrège et al. (2004) report seasonal salinity minima occurring in winter during the Younger Dryas in New Caledonia instead of summer as it does today. Further, the approach described here assumes linear relationships and covariance among variables, assumptions that may break down at some sites (e.g., where extreme precipitation drives a non-linear relationship between SST and SSS). Nevertheless, our results demonstrate the important contribution of $\delta^{18}O_{sw}$ variability to $\delta^{18}O_{coral}$ records from many sites—variability which may covary with local temperature in complex (non-linear, non-stationary, or destructive) ways; therefore $\delta^{18}O_{sw}$ variability cannot be ignored in $\delta^{18}O_{coral}$ -based temperature or salinity reconstructions.

The F_{sw} values do not vary appreciably on interannual to multidecadal timescales relative to the entire 1950–2009 period. Importantly, sites with high F_{sw} values maintain high F_{sw} values in different frequency bandwidths, indicating that hydro-sensitive coral sites should preserve salinity information across timescales. Detrended and raw F_{sw} values are also broadly similar, though slightly higher F_{sw} values are obtained with detrended time series in some regions, particularly the subtropical Pacific and throughout the Indian Ocean. Nevertheless, the difference is small, and our results suggest that on the interannual to multidecadal timescales, SSS information in $\delta^{18}O_{coral}$ values should remain consistent in sites where F_{sw} is high.

The regions where a large portion of coral δ^{18} O variance is driven by $\delta^{18}O_{sw}$ are areas of convergence and intense convection, resulting in high rates of regional precipitation. As a result, SSS and $\delta^{18}O_{sw}$ values are notably lower in these areas (LeGrande & Schmidt, 2006). Local a_2 coefficients are also relatively high (0.3–0.4‰/PSU), compared to the tropical Pacific mean of 0.27‰/PSU and eastern tropical Pacific value ~0.1 ‰/PSU (Benway & Mix, 2004; Conroy et al., 2017). High a_2 coefficients in the western tropical Pacific are hypothesized to be a result of the lower δ^{18} O value of precipitation, a result of the isotope "amount effect," a complex phenomenon that is the result of moisture recycling and rain re-evaporation in storms, among other processes (Conroy et al., 2016; Dansgaard, 1964; Risi et al., 2008).

The coefficient a_2 plays a key role in setting F_{sw} , and higher a_2 coefficients likely amplify the inferred $\delta^{18}O_{sw}$ -SSS signal in $\delta^{18}O_{coral}$ values in the western tropical Pacific. Future work should prioritize seawater $\delta^{18}O_{sw}$ -SSS monitoring to better constrain the a_2 coefficient and thus the inferred hydroclimate signal within these $\delta^{18}O_{coral}$ records. Monitoring efforts should include long time series from key reef sites to assess the space-for-time assumption often made in paleoclimate studies (i.e., that the spatial relationship across sites is representative of the relationship through time at individual sites). Such data could be obtained by installing networks of OsmoSamplers, which do not require power and can be deployed for up to a year collecting water samples on daily to monthly time intervals (Jannasch et al., 2004; Khare et al., 2021), or by making continuous shipboard measurements with spectroscopic water isotope analyzers fed seawater from the intake line (Munksgaard et al., 2012). In each case, precise SSS measurements must also be obtained. Furthermore, there is a critical need for an updated, actively maintained, and community-based seawater isotope database where laboratories can upload $\delta^{18}O_{sw}$ data.

Circumventing this uncertainty by modeling $\delta^{18}O_{coral}$ directly from $\delta^{18}O_{sw}$ simulated from isoROMS and iCESM, we demonstrate that the spatial pattern of F_{sw} is consistent with that of observational and reanalysis-based estimates, emphasizing the utility of this approach in identifying hydro-sensitive coral sites. Nevertheless, we find that the values are notably weaker in iCESM due to weak $\delta^{18}O_{sw}$ variability and F_{cov} (Figure S13 in Supporting Information S1). These results imply that iCESM underestimates $\delta^{18}O_{sw}$ variability and its covariance with SST, or that the uncertainty in a_2 causes an overestimation of F_{sw} when calculated from observational SSS. The strong correspondence between isoROMS and observational estimates suggests that these differences are likely attributable to biases in iCESM, particularly in the WPWP where iCESM shows negative F_{cov} values (i.e., a positive SST-SSS/ $\delta^{18}O_{sw}$ relationship). Nonetheless, these modeling approaches are necessary in order to fully quantify the complex relationships between SST and $\delta^{18}O_{sw}$ noted above, and eventually to understand the dynamics responsible for their variation.

This work highlights sites where $\delta^{18}O_{coral}$ values can provide critical insights into SSS, and thus tropical hydroclimate variability and change. It also emphasizes the importance of site selection and screening methods to identify hydro-sensitive sites where the signal is not dampened by SST variability. Many hydro-sensitive coral records identified have long records that predate the beginning of reliable SSS data (Figure 4), and thus can provide much needed context on the anomalous nature of late-20th century SSS trends. Indeed, all 1900–2009 trends in the hydro-sensitive IPWP records are negative, and thus a sizable portion of these trends is likely reflecting ocean freshening, rather than warming alone. These results are consistent with historical data (e.g., Durack et al., 2012), and suggest an intensification of the hydrological cycle and/or the Pacific Walker circulation.

At such sites where $\delta^{18}O_{sw}$ contributes a significant portion of the $\delta^{18}O_{coral}$ variability (e.g., $F_{sw} > 0.2$), independent Sr/Ca-SST estimates are necessary for reconstructing temperature, as SSS may constructively or destructively contribute to the inferred $\delta^{18}O_{coral}$ -SST values in these locations. Although constructive SST-SSS covariance may be leveraged to assess changes in climate variability (e.g., associated with ENSO or IOD) to natural and anthropogenic forcing, the spatially varying importance of $\delta^{18}O_{sw}$ (i.e., F_{sw}) and its covariance with SST (i.e., F_{cov}) complicate the interpretation of variance estimates among sites (e.g., among atolls in the Line Islands, see Grothe et al., 2020).

The $\delta^{18}O_{coral}$ records from the hydro-sensitive coral sites identified here hold tremendous potential for SSS reconstruction. In sites where F_{sw} is high (>0.7–0.8), $\delta^{18}O_{coral}$ variations can be confidently attributed to $\delta^{18}O_{sw}$ and used as evidence of salinity change. Future data synthesis and sample collection efforts should prioritize these regions for robust hydroclimate reconstructions. In sites where F_{sw} is positive but not as high, $\delta^{18}O_{sw}$ and SST both contribute to the coral record and cannot be simply separated. The framework presented here therefore provides a means to quantify the relative influence of temperature and salinity on $\delta^{18}O_{coral}$ values, improving both temperature and salinity reconstructions from these critical high-resolution climate archives.

Data Availability Statement

There was no new data generated in this study. Coral data can be accessed through the Iso2k database at: https:// www.ncei.noaa.gov/metadata/geoportal/rest/metadata/item/noaa-recon-29593/html, Coral metadata (both previously published and calculated in this study) can be found in Table S1 in Supporting Information S1 and on the NCEI paleoclimate database (https://www.ncei.noaa.gov/access/paleo-search/study/36273); the code is available at 10.5281/zenodo.6450406.

Acknowledgments

The authors acknowledge Iso2k, a contribution to Phase 3 of the PAGES 2k Network, for the database used in this manuscript and Hussein Savani and Emilie Dassie for their role in curating and quality controlling data as part of the coral archive team. PAGES received support from the Swiss Academy of Sciences, the US National Science Foundation, and the Chinese Academy of Sciences. Support for this work comes from NSF-CA-REER-1945479 and NSF-1931242 to DMT. NSF-CAREER-1847791 to JLC. NSF-AGS-1805141 to BLK, NSF-AGS-PRF-1433408 to BLK, NSF-1805143 to SS, NSF-1805702, NSF-2102931 and South Central Climate Adaptation Science Center Cooperative Agreement G19AC00086 for KLD, and the PALMOD climate modeling initiative supported by the German Ministry of Science and Education for LJ.

References

- Behringer, D., & Xue, Y. (2004). Evaluation of the global ocean data assimilation system at NCEP: The Pacific Ocean. In Proc. eighth symp. on integrated observing and assimilation systems for atmosphere, oceans, and land surface.
- Benway, H. M., & Mix, A. C. (2004). Oxygen isotopes, upper-ocean salinity, and precipitation sources in the eastern tropical Pacific. *Earth and Planetary Science Letters*, 224(3–4), 493–507. https://doi.org/10.1016/j.epsl.2004.05.014
- Cahyarini, S. Y., Pfeiffer, M., Nurhati, I. S., Aldrian, E., Dullo, W. C., & Hetzinger, S. (2014). Twentieth century sea surface temperature and salinity variations at Timor inferred from paired coral δ¹⁸O and Sr/Ca measurements. *Journal of Geophysical Research: Oceans*, 119(7), 4593–4604. https://doi.org/10.1002/2013jc009594
- Cahyarini, S. Y., Pfeiffer, M., Timm, O., Dullo, W.-C., & Schönberg, D. G. (2008). Reconstructing seawater δ¹⁸O from paired coral δ¹⁸O and Sr/ Ca ratios: Methods, error analysis and problems, with examples from Tahiti (French Polynesia) and Timor (Indonesia). *Geochimica et Cosmochimica Acta*, 72(12), 2841–2853. https://doi.org/10.1016/j.gca.2008.04.005
- Carré, M., Braconnot, P., Elliot, M., d'Agostino, R., Schurer, A., Shi, X., et al. (2021). High-resolution marine data and transient simulations support orbital forcing of ENSO amplitude since the mid-Holocene. *Quaternary Science Reviews*, 268, 107125. https://doi.org/10.1016/j. quascirev.2021.107125
- Carton, J. A., & Giese, B. (2008). A reanalysis of ocean climate using simple ocean data assimilation (SODA). *Monthly Weather Review*, 136(8), 2999–3017. https://doi.org/10.1175/2007mwr1978.1
- Cheng, L., Trenberth, K. E., Gruber, N., Abraham, J. P., Fasullo, J. T., Li, G., et al. (2020). Improved estimates of changes in upper ocean salinity and the hydrological cycle. *Journal of Climate*, 33(23), 10357–10381. https://doi.org/10.1175/JCLI-D-20-0366.1
- Cobb, K. M., Westphal, N., Sayani, H. R., Watson, J. T., Di Lorenzo, E., Cheng, H., et al. (2013). Highly variable El Niño-Southern Oscillation throughout the Holocene. *Science*, 339(6115), 67–70. https://doi.org/10.1126/science.1228246
- Cole, J. E., & Fairbanks, R. G. (1990). The southern oscillation recorded in the δ¹⁸O of corals from tarawa atoll. *Paleoceanography*, 5, 669–683. https://doi.org/10.1029/pa005i005p00669
- Conroy, J. L., Noone, D., Cobb, K. M., Moerman, J. W., & Konecky, B. L. (2016). Paired stable isotopologues in precipitation and vapor: A case study of the amount effect within Western tropical Pacific storms. *Journal of Geophysical Research: Atmospheres*, 121(7), 3290–3303. https:// doi.org/10.1002/2015jd023844
- Conroy, J. L., Thompson, D. M., Cobb, K. M., Noone, D., Rea, S., & Legrande, A. N. (2017). Spatiotemporal variability in the δ¹⁸O-salinity relationship of seawater across the tropical Pacific Ocean. *Paleoceanography*, *32*(5), 484–497. https://doi.org/10.1002/2016pa003073
- Corrège, T. (2006). Sea surface temperature and salinity reconstruction from coral geochemical tracers. *Palaeogeography, Palaeoclimatology, Palaeoecology, 232*(2–4), 408–428. https://doi.org/10.1016/j.palaeo.2005.10.014
- Corrège, T., Gagan, M. K., Beck, J. W., Burr, G. S., Cabioch, G., & Le Cornec, F. (2004). Interdecadal variation in the extent of South Pacific tropical waters during the Younger Dryas event. *Nature*, 428(6986), 927–929. https://doi.org/10.1038/nature02506
- Craig, H., & Gordon, L. I. (1965). Deuterium and oxygen-18 variations in the ocean and the marine environment. In E. Tongiorgi(Ed.), *Stable isotopes in oceanographic studies and paleotemperatures* (pp. 9–130).

Dansgaard, W. (1964). Stable isotopes in precipitation. Tellus, 16(4), 436-468. https://doi.org/10.1111/j.2153-3490.1964.tb00181.x

Dassié, E. P., Hasson, A., Khodri, M., Lebas, N., & Linsley, B. K. (2018). Spatiotemporal variability of the south Pacific convergence zone fresh pool eastern from coral-derived surface salinity data. *Journal of Climate*, 31(8), 3265–3288. https://doi.org/10.1175/jcli-d-17-0071.1 Delcroix, T., Alory, G., Cravatte, S., Corrège, T., & McPhaden, M. (2011). A gridded sea surface salinity data set for the tropical Pacific with sample applications (1950–2008). Deep Sea Research Part I: Oceanographic Research Papers, 58(1), 38–48. https://doi.org/10.1016/j. dsr.2010.11.002

Durack, P. J., Wijffels, S. E., & Matear, R. J. (2012). Ocean salinities reveal strong global water cycle intensification during 1950-2000. *Science*, 336(6080), 455–458. https://doi.org/10.1126/science.1212222

Emiliani, C., Hudson, J. H., Shinn, E. A., & George, R. Y. (1978). Oxygen and carbon isotopic growth record in a reef coral from the Florida Keys and a deep-sea coral from Blake Plateau. Science, 202(4368), 627–629. https://doi.org/10.1126/science.202.4368.627

Epstein, S., Buchsbaum, R., Lowenstam, H. A., & Urey, H. C. (1953). Revised carbonate-water isotopic temperature scale. *The Geological Society of America Bulletin*, 64(11), 1315–1326. https://doi.org/10.1130/0016-7606(1953)64[1315:rcits]2.0.co;2

- Evans, M. N., Kaplan, A., & Cane, M. A. (2000). Intercomparison of coral oxygen isotope data and historical sea surface temperature (SST): Potential for coral-based SST field reconstructions. *Paleoceanography*, 15(5), 551–563. https://doi.org/10.1029/2000pa000498
- Fairbanks, R. G., Evans, M. N., Rubenstone, J. L., Mortlock, R. A., Broad, K., Moore, M. D., & Charles, C. D. (1997). Evaluating climate indices and their geochemical proxies measured in corals. *Coral Reefs*, 16(5), S93–S100. https://doi.org/10.1007/s003380050245
- Giese, B. S., & Ray, S. (2011). El Niño variability in simple ocean data assimilation (SODA), 1871-2008. Journal of Geophysical Research, 116(2), 1–17. https://doi.org/10.1029/2010JC006695
- Gorman, M. K., Quinn, T. M., Taylor, F. W., Partin, J. W., Cabioch, G., Austin, J. A., Jr. et al. (2012). A coral-based reconstruction of sea surface salinity at Sabine Bank, Vanuatu from 1842 to 2007 CE. *Paleoceanography*, 27.
- Grothe, P. R., Cobb, K. M., Liguori, G., Di Lorenzo, E., Capotondi, A., Lu, Y., et al. (2020). Enhanced El Niño–Southern oscillation variability in recent decades. *Geophysical Research Letters*, 47(7), e2019GL083906. https://doi.org/10.1029/2019gl083906
- Hereid, K. A., Quinn, T. M., Taylor, F. W., Shen, C. C., Edwards, R. L., & Cheng, H. (2013). Coral record of reduced El Niño activity in the early 15th to middle 17th centuries. *Geology*, 41(1), 51–54. https://doi.org/10.1130/g33510.1
- Jannasch, H. W., Wheat, C. G., Plant, J. N., Kastner, M., & Stakes, D. S. (2004). Continuous chemical monitoring with osmotically pumped water samplers: OsmoSampler design and applications. *Limnology and Oceanography: Methods*, 2(4), 102–113. https://doi.org/10.4319/ lom.2004.2.102
- Khare, A., Hughes, H. P., Kilbourne, K. H., & Schijf, J. (2021). An ICP-AES method for routine high-precision measurement of seawater Sr/ Ca ratios to validate coral paleothermometry calibrations. *Limnology and Oceanography: Methods*, 19(6), 416–430. https://doi.org/10.1002/ lom3.10434
- Konecky, B. L., & McKay, N. P. (2020). NOAA/WDS paleoclimatology The Iso2k database. NOAA national centers for environmental information. https://doi.org/10.25921/57j8-vs18
- Konecky, B. L., McKay, N. P., Comas-Bru, L., Dassié, E. P., DeLong, K. L., Falster, G. M., et al. (2020). The Iso2k database: A global compilation of paleo-δ¹⁸O and δ²H records to aid understanding of common era climate. *Earth System Science Data*, 12(3), 2261–2288. https://doi. org/10.5194/essd-12-2261-2020
- Lawman, A. E., Quinn, T. M., Partin, J. W., Thirumalai, K., Taylor, F. W., Wu, C. C., et al. (2020). A century of reduced ENSO variability during the Medieval Climate Anomaly. *Paleoceanography and Paleoclimatology*, 35(4), e2019PA003742. https://doi.org/10.1029/2019pa003742
- LeGrande, A. N., & Schmidt, G. A. (2006). Global gridded data set of the oxygen isotopic composition in seawater. *Geophysical Research Letters*, 33(12), L12604. https://doi.org/10.1029/2006gl026011
- LeGrande, A. N., & Schmidt, G. A. (2011). Water isotopologues as a quantitative paleosalinity proxy. Paleoceanography, 26(3).
- Lough, J. (2004). A strategy to improve the contribution of coral data to high-resolution paleoclimatology. *Palaeogeography, Palaeoclimatology, Palaeoecology, 204*(1–2), 115–143. https://doi.org/10.1016/s0031-0182(03)00727-2
- McGregor, H. V., Fischer, M. J., Gagan, M. K., Fink, D., Phipps, S. J., Wong, H., & Woodroffe, C. D. (2013). A weak El Niño/Southern Oscillation with delayed seasonal growth around 4,300 years ago. *Nature Geoscience*, 6(11), 949–953, https://doi.org/10.1038/NGEO1936
- Munksgaard, N. C., Wurster, C. M., Bass, A., Zagorskis, I., & Bird, M. I. (2012). First continuous shipboard delta O-18 and delta D measurements in sea water by diffusion sampling-cavity ring-down spectrometry. *Environmental Chemistry Letters*, 10(3), 301–307. https://doi.org/10.1007/ s10311-012-0371-5
- Murty, S. A., Goodkin, N. F., Wiguna, A. A., & Gordon, A. L. (2018). Variability in coral-reconstructed sea surface salinity between the northern and southern lombok strait linked to east Asian winter monsoon mean state reversals. *Paleoceanography and Paleoclimatology*, 33(10), 1116–1133. https://doi.org/10.1029/2018PA003387
- Nusbaumer, J., Wong, T. E., Bardeen, C., & Noone, D. (2017). Evaluating hydrological processes in the Community Atmosphere Model Version 5 (CAM5) using stable isotope ratios of water. *Journal of Advances in Modeling Earth Systems*, 9(2), 949–977. https://doi. org/10.1002/2016ms000839
- Rayner, N. A., Parker, D. E., Horton, E. B., Folland, C. K., Alexander, L. V., Rowell, D. P., et al. (2003). Global analyses of sea surface temperature, sea ice, and night marine air temperature since the late nineteenth century. *Journal of Geophysical Research*, 108(D14), 4407. https:// doi.org/10.1029/2002JD002670
- Reynolds, R. W., Rayner, N. A., Smith, T. M., Stokes, D. C., & Wang, W. Q. (2002). An improved in situ and satellite SST analysis for climate. Journal of Climate, 15(13), 1609–1625. https://doi.org/10.1175/1520-0442(2002)015<1609:aiisas>2.0.co;2
- Risi, C., Bony, S., & Vimeux, F. (2008). Influence of convective processes on the isotopic composition (δ¹⁸O and δD) of precipitation and water vapor in the tropics: 2. Physical interpretation of the amount effect. *Journal of Geophysical Research: Atmospheres*, 113(D19), D19306. https://doi.org/10.1029/2008jd009943
- Russon, T., Tudhope, A., Hegerl, G., Collins, M., & Tindall, J. (2013). Inter-annual tropical Pacific climate variability in an isotope-enabled CGCM: Implications for interpreting coral stable oxygen isotope records of ENSO. *Climate of the Past*, 9(4), 1543–1557. https://doi. org/10.5194/cp-9-1543-2013
- Stevenson, S., Powell, B. S., Cobb, K. M., Nusbaumer, J., Merrifield, M. A., & Noone, D. (2018). 20th century seawater δ¹⁸O dynamics and implications for coral-based climate reconstruction. *Paleoceanography and Paleoclimatology*, 33.
- Stevenson, S., Powell, B. S., Merrifield, M. A., Cobb, K. M., Nusbaumer, J., & Noone, D. (2015). Characterizing seawater oxygen isotopic variability in a regional ocean modeling framework: Implications for coral proxy records. *Paleoceanography*, 30(11), 1573–1593. https://doi. org/10.1002/2015pa002824
- Thompson, D. M., Ault, T. R., Evans, M. N., Cole, J. E., & Emile-Geay, J. (2011). Comparison of observed and simulated tropical climate trends using a forward model of coral δ¹⁸O. *Geophysical Research Letters*, 38.
- Tierney, J. E., Abram, N. J., Anchukaitis, K. J., Evans, M. N., Giry, C., Kilbourne, K. H., et al. (2015). Tropical sea surface temperatures for the past four centuries reconstructed from coral archives. *Paleoceanography*, 30(3), 226–252. https://doi.org/10.1002/2014pa002717
- Watanabe, T. K., Watanabe, T., Yamazaki, A., Pfeiffer, M., & Claereboudt, M. R. (2019). Oman coral δ^{18} O seawater record suggests that Western Indian Ocean upwelling uncouples from the Indian Ocean Dipole during the global-warming hiatus. *Scientific Reports*, 9(1), 1–9.

- Weber, J. N., & Woodhead, P. M. (1972). Temperature dependence of oxygen-18 concentration in reef coral carbonates. *Journal of Geophysical Research*, 77(3), 463–473. https://doi.org/10.1029/jc077i003p00463
- Zhang, J., Liu, Z., Brady, E. C., Oppo, D. W., Clark, P. U., Jahn, A., et al. (2017). Asynchronous warming and δ¹⁸O evolution of deep Atlantic water masses during the last deglaciation. *Proceedings of the National Academy of Sciences*, 114(42), 11075–11080. https://doi.org/10.1073/ pnas.1704512114
- Zhu, J., Liu, Z., Brady, E., Otto-Bliesner, B., Zhang, J., Noone, D., et al. (2017). Reduced ENSO variability at the LGM revealed by an isotope-enabled Earth system model. *Geophysical Research Letters*, 44(13), 6984–6992. https://doi.org/10.1002/2017g1073406
- Zuo, H., Balmaseda, M. A., Tietsche, S., Mogensen, K., & Mayer, M. (2019). The ECMWF operational ensemble reanalysis-analysis system for Ocean and sea ice: A description of the system and assessment. Ocean Science, 15(3), 779–808. https://doi.org/10.5194/os-15-779-2019

References From the Supporting Information

- Abram, N. J., Gagan, M. K., Cole, J. E., Hantoro, W. S., & Mudelsee, M. (2008). Recent intensification of tropical climate variability in the Indian Ocean. Nature Geoscience, 1(12), 849–853. https://doi.org/10.1038/ngeo357
- Abram, N. J., Dixon, B. C., Rosevear, M. G., Plunkett, B., Gagan, M. K., Hantoro, W. S., & Phipps, S. J. (2015). Optimized coral reconstructions of the Indian Ocean Dipole: An assessment of location and length considerations. *Paleoceanography*, 30(10), 1391–1405. https://doi. org/10.1002/2015pa002810
- Asami, R., Yamada, T., Iryu, Y., Quinn, T. M., Meyer, C. P., & Paulay, G. (2005). Interannual and decadal variability of the Western Pacific sea surface condition for the years 1787–2000: Reconstruction based on stable isotope record from a Guam coral. *Journal of Geophysical Research*, (C5), 110. https://doi.org/10.1029/2004jc002555
- Bagnato, S., Linsley, B. K., Howe, S. S., Wellington, G. M., & Salinger, J. (2004). Evaluating the use of the massive coral *Diploastrea heliopora* for paleoclimate reconstruction. *Paleoceanography*, 19(1). https://doi.org/10.1029/2003pa000935
- Boiseau, M., Juillet-Leclerc, A., Yiou, P., Salvat, B., Isdale, P., & Guillaume, M. (1998). Atmospheric and oceanic evidences of El Niño-Southern Oscillation events in the south central Pacific Ocean from coral stable isotopic records over the last 137 years. *Paleoceanography*, 13(6), 671–685. https://doi.org/10.1029/98pa02502
- Calvo, E., Marshall, J. F., Pelejero, C., McCulloch, M. T., Gagan, M. K., & Lough, J. M. (2007). Interdecadal climate variability in the coral sea since 1708 AD. Palaeogeography, Palaeoclimatology, Palaeoecology, 248(1–2), 190–201. https://doi.org/10.1016/j.palaeo.2006.12.003
- Carilli, J., Charles, C., Garren, M., McField, M., & Norris, R. (2013). Baseline shifts in coral skeletal oxygen isotopic composition: A signature of symbiont shuffling? *Coral Reefs*, 32(2), 559–571. https://doi.org/10.1007/s00338-012-1004-y
- Carilli, J. E., McGregor, H. V., Gaudry, J. J., Donner, S. D., Gagan, M. K., Stevenson, S., et al. (2014). Equatorial Pacific coral geochemical records show recent weakening of the Walker Circulation. *Paleoceanography*, 29(11), 1031–1045. https://doi.org/10.1002/2014pa002683
- Chakraborty, S., & Ramesh, R. (1998). Stable isotope variations in a coral (*Favia speciosa*) from the Gulf of Kutch during 1948–1989 AD: Environmental implications. *Proceedings of the Indian Academy of Sciences - Earth & Planetary Sciences*, 107(4), 331–341. https://doi. org/10.1007/bf02841599
- Charles, C. D., Hunter, D. E., & Fairbanks, R. G. (1997). Interaction between the ENSO and the Asian monsoon in a coral record of tropical climate. Science, 277(5328), 925–928. https://doi.org/10.1126/science.277.5328.925
- Charles, C. D., Cobb, K. M., Moore, M. D., & Fairbanks, R. G. (2003). Monsoon-tropical ocean interaction in a network of coral records spanning the 20th century. *Marine Geology*, 201(1–3), 207–222. https://doi.org/10.1016/s0025-3227(03)00217-2
- Cobb, K. M., Charles, C. D., Cheng, H., & Edwards, R. L. (2003). El Niño/Southern Oscillation and tropical Pacific climate during the last millennium. Nature, 424(6946), 271–276. https://doi.org/10.1038/nature01779
- Cole, J. E., Fairbanks, R. G., & Shen, G. T. (1993). Recent variability in the southern oscillation Isotopic results from a tarawa atoll coral. *Science*, 260(5115), 1790–1793. https://doi.org/10.1126/science.260.5115.1790
- Cole, J. E., Dunbar, R. B., McClanahan, T. R., & Muthiga, N. A. (2000). Tropical Pacific forcing of decadal SST variability in the Western Indian Ocean over the past two centuries. *Science*, 287(5453), 617–619. https://doi.org/10.1126/science.287.5453.617
- Damassa, T. D., Cole, J. E., Barnett, H. R., Ault, T. R., & McClanahan, T. R. (2006). Enhanced multidecadal climate variability in the seventeenth century from coral isotope records in the Western Indian Ocean. *Paleoceanography*, 21(2). https://doi.org/10.1029/2005PA001217
- Deng, W., Wei, G., Xie, L., Ke, T., Wang, Z., Zeng, T., & Liu, Y. (2013). Variations in the Pacific decadal oscillation since 1853 in a coral record from the northern South China Sea. Journal of Geophysical Research: Oceans, 118(5), 2358–2366. https://doi.org/10.1002/jgrc.20180
- Druffel, E. R. M., & Griffin, S. (1999). Variability of surface ocean radiocarbon and stable isotopes in the southwestern Pacific. Journal of Geophysical Research-Oceans, 104(C10), 23607–23613. https://doi.org/10.1029/1999jc900212
- Dunbar, R. B., Wellington, G. M., Colgan, M. W., & Glynn, P. W. (1994). Eastern Pacific SST since 1600 AD the 8¹⁸O record of climate variability in Galápagos corals. *Paleoceanography*, 9(2), 291–315. https://doi.org/10.1029/93pa03501
- Evans, M. N., Fairbanks, R. G., & Rubenstone, J. L. (1998). A proxy for ENSO teleconnections. *Nature*, 394(6695), 732–733. https://doi.org/10.1038/29424
- Felis, T., Pätzold, J., Loya, Y., Fine, M., Nawar, A. H., & Wefer, G. (2000). A coral oxygen isotope record from the northern Red Sea documenting NAO, ENSO, and North Pacific teleconnections on Middle East climate variability since the year 1750. *Paleoceanography*, 15(6), 679–694. https://doi.org/10.1029/1999pa000477
- Felis, T., Suzuki, A., Kuhnert, H., Dima, M., Lohmann, G., & Kawahata, H. (2009). Subtropical coral reveals abrupt early-twentieth-century freshening in the Western North Pacific Ocean. *Geology*, 37(6), 527–530. https://doi.org/10.1130/g25581a.1
- Goodkin, N. F., Hughen, K. A., Doney, S. C., & Curry, W. B. (2008). Increased multidecadal variability of the North Atlantic Oscillation since 1781. Nature Geoscience, 1(12), 844–848. https://doi.org/10.1038/ngeo352
- Guilderson, T. P., & Schrag, D. P. (1999). Reliability of coral isotope records from the Western Pacific warm pool: A comparison using age-optimized records. *Paleoceanography*, 14(4), 457–464. https://doi.org/10.1029/1999pa900024
- Heiss, G. A. (1994). Coral reefs in the Red Sea: Growth, production and stable isotopes. *GEOMAR-Report*, 32, 1–141.
- Hennekam, R., Zinke, J., van Sebille, E., ten Have, M., Brummer, G. J. A., & Reichart, G. J. (2018). Cocos (Keeling) corals reveal 200 years of multidecadal modulation of southeast Indian Ocean hydrology by Indonesian throughflow. *Paleoceanography and Paleoclimatology*, 33(1), 48–60. https://doi.org/10.1002/2017pa003181
- Hetzinger, S., Pfeiffer, M., Dullo, W.-C., Keenlyside, N., Latif, M., & Zinke, J. (2008). Caribbean coral tracks Atlantic Multidecadal Oscillation and past hurricane activity. *Geology*, 36(1), 11–14. https://doi.org/10.1130/g24321a.1

- Hetzinger, S., Pfeiffer, M., Dullo, W.-C., Garbe-Schönberg, D., & Halfar, J. (2010). Rapid 20th century warming in the Caribbean and impact of remote forcing on climate in the northern tropical Atlantic as recorded in a Guadeloupe coral. *Palaeogeography, Palaeoclimatology, Palaeoeology, 296*(1–2), 111–124. https://doi.org/10.1016/j.palaeo.2010.06.019
- Kilbourne, K. H., Quinn, T. M., Taylor, F. W., Delcroix, T., & Gouriou, Y. (2004). El Niño–Southern Oscillation–related salinity variations recorded in the skeletal geochemistry of a Porites coral from Espiritu Santo, Vanuatu. *Paleoceanography*, 19(4). https://doi.org/10.1029/2004pa001033
- Kilbourne, K., Quinn, T., Webb, R., Guilderson, T., Nyberg, J., & Winter, A. (2008). Paleoclimate proxy perspective on Caribbean climate since the year 1751: Evidence of cooler temperatures and multidecadal variability. *Paleoceanography*, 23(3). https://doi.org/10.1029/2008pa001598
- Kuhnert, H., Pätzold, J., Wyrwoll, K.-H., & Wefer, G. (2000). Monitoring climate variability over the past 116 years in coral oxygen isotopes from Ningaloo Reef, Western Australia. *International Journal of Earth Sciences*, 88(4), 725–732. https://doi.org/10.1007/s005310050300
- Kuhnert, H., Crüger, T., & Pätzold, J. (2005). NAO signature in a Bermuda coral Sr/Ca record. *Geochemistry, Geophysics, Geosystems*, 6(4). https://doi.org/10.1029/2004gc000786
- Kuhnert, H., Pätzold, J., Hatcher, B., Wyrwoll, K.-H., Eisenhauer, A., Collins, L., et al. (1999). A 200-year coral stable oxygen isotope record from a high-latitude reef off Western Australia. Coral Reefs, 18, 1–12. https://doi.org/10.1007/s003380050147
- Li, X., Liu, Y., Hsin, Y. C., Liu, W., Shi, Z., Chiang, H. W., & Shen, C. C. (2017). Coral record of variability in the upstream Kuroshio Current during 1953–2004. Journal of Geophysical Research: Oceans, 122(8), 6936–6946. https://doi.org/10.1002/2017jc012944
- Linsley, B., Messier, R., & Dunbar, R. (1999). Assessing between-colony oxygen isotope variability in the coral *Porites lobata* at Clipperton Atoll. *Coral Reefs*, 18(1), 13–27. https://doi.org/10.1007/s003380050148
- Linsley, B., Wellington, G., Schrag, D., Salinger, M. J., Ren, L., Linsley, B. K., et al. (2004). Geochemical evidence from corals for changes in the amplitude and spatial pattern of South Pacific interdecadal climate variability over the last 300 years. *Climate Dynamics*, 22, 1–11. https:// doi.org/10.1007/s00382-003-0364-y
- Linsley, B. K., Kaplan, A., Gouriou, Y., Salinger, J., Demenocal, P. B., Wellington, G. M., & Howe, S. S. (2006). Tracking the extent of the South Pacific Convergence Zone since the early 1600s. *Geochemistry, Geophysics, Geosystems*, 7(5). https://doi.org/10.1029/2005gc001115
- Linsley, B. K., Dunbar, R. B., Wellington, G. M., & Mucciarone, D. A. (1994). A coral-based reconstruction of Intertropical Convergence Zone variability over Central America since 1707. Journal of Geophysical Research, 99(C5), 9977–9994. https://doi.org/10.1029/94jc00360
- McGregor, H. V., & Gagan, M. K. (2004). Western Pacific coral δ¹⁸O records of anomalous Holocene variability in the El Niño-Southern Oscillation. *Geophysical Research Letters*, 31(11). https://doi.org/10.1029/2004gl019972
- Moses, C. S., Swart, P. K., & Rosenheim, B. E. (2006). Evidence of multidecadal salinity variability in the eastern tropical North Atlantic. Paleoceanography, 21(3). https://doi.org/10.1029/2005pa001257
- Moses, C. S., & Swart, P. K. (2006). Stable isotope and growth records in corals from the island of Tobago: Not simply a record of the Orinoco (Okinawa). In *Proceedings of the 10th international coral reef symposium* (pp. 580–587).
- Murty, S., Goodkin, N., Halide, H., Natawidjaja, D., Suwargadi, B., Suprihanto, I., et al. (2017). Climatic influences on southern Makassar Strait salinity over the past century. *Geophysical Research Letters*, 44(23), 11967–11975. https://doi.org/10.1002/2017gl075504
- Nakamura, N., Kayanne, H., Iijima, H., McClanahan, T. R., Behera, S. K., & Yamagata, T. (2009). Mode shift in the Indian Ocean climate under global warming stress. *Geophysical Research Letters*, 36(23), L23708. https://doi.org/10.1029/2009gl040590
- Nurhati, I. S., Cobb, K. M., & Di Lorenzo, E. (2011). Decadal-scale SST and salinity variations in the central tropical Pacific: Signatures of natural and anthropogenic climate change. *Journal of Climate*, 24(13), 3294–3308. https://doi.org/10.1175/2011jcli3852.1
- Osborne, M. C., Dunbar, R. B., Mucciarone, D. A., Druffel, E., & Sanchez-Cabeza, J.-A. (2014). A 215-yr coral δ^{18} O time series from Palau records dynamics of the West Pacific Warm Pool following the end of the Little Ice Age. *Coral Reefs*, 33(3), 719–731. https://doi.org/10.1007/s00338-014-1146-1
- Pfeiffer, M., Dullo, W.-C., & Eisenhauer, A. (2004). Variability of the intertropical convergence zone recorded in coral isotopic records from the central Indian Ocean (Chagos Archipelago). *Quaternary Research*, 61(3), 245–255. https://doi.org/10.1016/j.yqres.2004.02.009
- Quinn, T. M., Crowley, T. J., & Taylor, F. W. (1996). New stable isotope results from a 173-year coral from Espiritu Santo, Vanuatu. Geophysical Research Letters, 23(23), 3413–3416. https://doi.org/10.1029/96g103169
- Quinn, T. M., Crowley, T. J., Taylor, F. W., Henin, C., Joannot, P., & Join, Y. (1998). A multicentury stable isotope record from a New Caledonia coral: Interannual and decadal sea surface temperature variability in the southwest Pacific since 1657 AD. *Paleoceanography*, 13(4), 412–426. https://doi.org/10.1029/98pa00401
- Quinn, T. M., Taylor, F. W., & Crowley, T. J. (2006). Coral-based climate variability in the Western Pacific Warm Pool since 1867. Journal of Geophysical Research, (C11), 111. https://doi.org/10.1029/2005jc003243
- Ramos, R. D., Goodkin, N. F., Siringan, F. P., & Hughen, K. A. (2017). Diploastrea heliopora Sr/Ca and δ¹⁸O records from northeast Luzon, Philippines: An assessment of interspecies coral proxy calibrations and climate controls of sea surface temperature and salinity. *Paleoceanography*, 32(4), 424–438. https://doi.org/10.1002/2017pa003098
- Shen, G. T., Cole, J. E., Lea, D. W., Linn, L. J., McConnaughey, T. A., & Fairbanks, R. G. (1992). Surface ocean variability at Galapagos from 1936-1982: Calibration of geochemical tracers in corals. *Paleoceanography*, 7(5), 563–588. https://doi.org/10.1029/92pa01825
- Swart, P. K., Dodge, R. E., & Hudson, H. J. (1996a). A 240-year stable oxygen and carbon isotopic record in a coral from South Florida: implications for the prediction of precipitation in southern Florida. PALAIOS, 11(4), 362–375. https://doi.org/10.2307/3515246
- Swart, P. K., Healy, G. F., Dodge, R. E., Kramer, P., Hudson, J. H., Halley, R. B., & Robblee, M. B. (1996b). The stable oxygen and carbon isotopic record from a coral growing in Florida bay: A 160 year record of climatic and anthropogenic influence. *Palaeogeography, Palaeocli*matology, *Palaeoecology*, 123(1–4), 219–237. https://doi.org/10.1016/0031-0182(95)00078-x
- Swart, P. K., White, K. S., Enfield, D., Dodge, R. E., & Milne, P. (1998). Stable oxygen isotopic composition of corals from the Gulf of Guinea as indicators of periods of extreme precipitation conditions in the sub-Sahara. *Journal of Geophysical Research*, 103(C12), 27885–27891. https://doi.org/10.1029/98jc02404
- Swart, P. K., Healy, G., Greer, L., Lutz, M., Saied, A., Anderegg, D., et al. (1999). The use of proxy chemical records in coral skeletons to ascertain past environmental conditions in Florida Bay. *Estuaries*, 22(2), 384–397. https://doi.org/10.2307/1353206
- Tudhope, A. W., Chilcott, C. P., McCulloch, M. T., Cook, E. R., Chappell, J., Ellam, R. M., et al. (2001). Variability in the El Niño-Southern Oscillation through a glacial-interglacial cycle. *Science*, 291(5508), 1511–1517. https://doi.org/10.1126/science.1057969
- Tudhope, A. W., Shimmield, G. B., Chilcott, C. P., Jebb, M., Fallick, A. E., & Dalgleish, A. N. (1995). Recent changes in climate in the far western equatorial Pacific and their relationship to the Southern Oscillation; oxygen isotope records from massive corals, Papua New Guinea. *Earth* and Planetary Science Letters, 136(3–4), 575–590. https://doi.org/10.1016/0012-821x(95)00156-7
- Urban, F. E., Cole, J. E., & Overpeck, J. T. (2000). Influence of mean climate change on climate variability from a 155-year tropical Pacific coral record. *Nature*, 407(6807), 989–993. https://doi.org/10.1038/35039597



- von Reumont, J., Hetzinger, S., Garbe-Schönberg, D., Manfrino, C., & Dullo, C. (2018). Tracking interannual-to multidecadal-scale climate variability in the Atlantic Warm Pool using central Caribbean coral data. *Paleoceanography and Paleoclimatology*, 33(4), 395–411. https:// doi.org/10.1002/2018pa003321
- Wu, H. C., Moreau, M., Linsley, B. K., Schrag, D. P., & Corrège, T. (2014). Investigation of sea surface temperature changes from replicated coral Sr/Ca variations in the eastern equatorial Pacific (Clipperton Atoll) since 1874. *Palaeogeography, Palaeoclimatology, Palaeoecology, 412*, 208–222. https://doi.org/10.1016/j.palaeo.2014.07.039
- Zinke, J., Dullo, W.-C., Heiss, G., & Eisenhauer, A. (2004). ENSO and Indian Ocean subtropical dipole variability is recorded in a coral record off southwest Madagascar for the period 1659 to 1995. *Earth and Planetary Science Letters*, 228(1–2), 177–194. https://doi.org/10.1016/j. epsl.2004.09.028
- Zinke, J., Rountrey, A., Feng, M., Xie, S.-P., Dissard, D., Rankenburg, K., et al. (2014). Corals record long-term Leeuwin current variability including Ningaloo Niño/Niña since 1795. Nature Communications, 5, 1–9. https://doi.org/10.1038/ncomms4607
- Zinke, J., Loveday, B. R., Reason, C., Dullo, W.-C., & Kroon, D. (2014). Madagascar corals track sea surface temperature variability in the Agulhas Current core region over the past 334 years. *Scientific Reports*, *4*, 1–8. https://doi.org/10.1038/srep04393